Author's response to reviews

Title: Contribution of germline BRCA1 and BRCA2 sequence alterations to breast cancer in Northern India.

Authors:

Sunita Saxena (sunita_saxena@yahoo.com)
Anurupa Chakraborty (anurupa_chakraborty@yahoo.co.in)
Sanjeev Kotwal (drsanjeevkotwal@gmail.com)
Dinesh Bhatnagar (dbhatnagar@yahoo.com)
RavindarShing Mohil (rsmohil@yahoo.com)
Chintamani Chintamani (chintamani7@rediffmail.com)
Ak Aggarwal (bhaganwal@sapient.com)
Veena Sharma (sunita_saxena@yahoo.com)
PrakashChand Sharma (sharmapc_meerut@yahoo.com)
Gilbert Lenior (lenoir@igr.fr)
David Goldgar (goldgar@iarc.fr)
Csilla Szabo (szabo_csilla@hotmail.com)

Version: 4 Date: 18 August 2006

Author's response to reviews:

Dear Editor

we have modified the manuscript as per suggestion. we want to add the name of Mishi Kaushal at the third position in author's list because she has done the HDA in the new controls added in the present study.

The revised manuscript is attached along with the reply of reviewer's comments.